# Finding a Path to Formalization in Benin

Early Results after the Introduction of the *Entreprenant* Legal Status

> Najy Benhassine David McKenzie Victor Pouliquen Massimiliano Santini





# Abstract

In April 2014, the Government of Benin launched the entreprenant status, a simplified and free legal regime offered to small informal businesses to enter the formal economy. This paper presents the short-term results of a randomized impact evaluation testing three different versions of the entreprenant status on business registration decisions, each version including incremental incentives to registration: (i) information on the new legal status and its benefits, (ii) business training, counseling services, and support to open a bank account, (iii) tax mediation services. The study included 3,600 informal businesses operating with a fixed location in Cotonou, Benin, which were randomly allocated between three treatment groups and one control group. One year after the program launch, all versions of the program had significant impact on formalization rates. The impact was 9.1 percentage points in the first treatment group; 13 percentage points in the second group; and 15.8 percentage points in the last group. The program had a higher impact on male business owners, with more education, operating outside Dantokpa Market, in sectors other than trade, and that before being offered the incentives to formalization had characteristics similar to businesses that were already formal. Data from a second follow-up survey, which is expected to take place in March 2016, will explore the impacts on other outcomes, like business performances or access to banking.

The Policy Research Working Paper Series disseminates the findings of work in progress to encourage the exchange of ideas about development issues. An objective of the series is to get the findings out quickly, even if the presentations are less than fully polished. The papers carry the names of the authors and should be cited accordingly. The findings, interpretations, and conclusions expressed in this paper are entirely those of the authors. They do not necessarily represent the views of the International Bank for Reconstruction and Development/World Bank and its affiliated organizations, or those of the Executive Directors of the World Bank or the governments they represent.

This paper is a product of the Trade and Competitiveness Global Practice Group. It is part of a larger effort by the World Bank to provide open access to its research and make a contribution to development policy discussions around the world. Policy Research Working Papers are also posted on the Web at http://econ.worldbank.org. The authors may be contacted at msantini@ifc.org.

# Finding a Path to Formalization in Benin: Early Results after the Introduction of the *Entreprenant* Legal Status<sup>1</sup>

Najy Benhassine, *World Bank Group* David McKenzie, *World Bank Group* Victor Pouliquen, *World Bank Group* Massimiliano Santini, *World Bank Group* 

<sup>&</sup>lt;sup>1</sup> The authors owe particular appreciation to all the experts and colleagues who contributed to the design and implementation of the *entreprenant* program and its impact evaluation: Louis Akakpo, Theodore Anthonioz, Zoubir Benhamouche, Julien Bornon, Laurent Corthay, Tonagnon Dadjo, Matina Deen, Magueye Dia, Benedicta Houetchenou, Ferdinand Ngobounan, Adrien Pawlik, Dolele Sylla, Kjartan Sorensen, Hamidou Sorgo, Adama Tiendrebeogo, and Alain Traore. The authors would also like to thank the Government of Benin for the cooperation and support during the impact evaluation, CGA (*Centres de Gestion Agrées*), GUFE (*Guichet Unique de Formalisation des Entreprises*), Bank of Africa, and Orabank. Research for this paper has been supported by the World Bank Group (WBG), the U.S. Agency for International Development, the U.K. Department for International Development (DFID), Foreign Affairs, Trade and Development Canada (DFATD), through the Impact Program managed by the WBG's Trade and Competitiveness Global Practice; by the WBG's Research Support Budget (RSB) and the Strategic Research Program (SRP); and by the Private Enterprise Development for Low-Income Countries (PEDL) initiative managed by the Centre for Economic Policy Research (CEPR) and DFID. For questions or comments, please contact Massimiliano Santini at <msantini@ifc.org>.

## 1. Introduction

In developing countries, a large majority of small and medium firms operate in the informal sector (La Porta and Shleifer, 2014a). In Benin, informality is particularly high: in 2009, the national statistics agency estimated that the informal sector represented up to 70 percent of the GDP and 95 percent of employment (INSAE, (2009)). High level of informality may carry some costs for both governments and firms. Governments may face more difficulties to collect taxes and finance public services (Levy 2008). Informal firms may not be able to access bank financing, public contracts and government programs and may face more corruption or intimidation from the authorities. Formal firms may suffer from unfair competition by the informal sector (Farrell (2004)) and from higher taxes if all the tax burden is supported by the formal sector. Two competing and not mutually exclusive theories may explain why most firms remain informal (Perry et al. (2007)). The exclusion hypothesis, often attributed to De Soto (1989), supports the view that administrative rigidities and direct costs of registration discourage firms from formalizing, whereas according to the *exit* hypothesis (Levy (2008), Maloney (2004)) firms rationally choose to stay informal by comparing costs and benefits of formality. Direct and indirect (i.e. added taxes) costs of registration may be greater than potential benefits for most informal businesses which are small, have very low productivity, and employ managers with a low level of human capital (La Porta and Shleifer, 2014b).

Following the publication of the *Doing Business* project of the World Bank, many countries around the world recently implemented a large number of business entry regulation reforms in order to reduce informality. For example, between 2004 and 2014, 100 of the 189 countries included in the *Doing Business* project implemented a "one-stop shop" for business registration to reduce the number of steps and the time required to register a firm (World Bank (2015)). However, the literature suggests two important results: (1) easing entry regulation and providing information may not be sufficient to increase formalization (at least when formality also means tax registration) and (2) formalization alone may not be sufficient in itself to increase firms' profits and stimulate growth.

A recent review of the literature by Bruhn and McKenzie (2014) suggests that reforms which only ease business registration have limited impact on formalization. In Mexico, Bruhn (2013) finds that a significant simplification of the registration process had a limited impact on informality. In Sri Lanka (de Mel et al. (2013)) and in Bangladesh (De Giorgi and Rahman (2013)), public programs providing information and cost reimbursement had no effect on the formalization rate. In Brazil, Andrade et al. (2013) evaluate the impact of three different interventions on business formalization. Providing information on the registration process and removing the cost of formalization had no effect on the formalization rate, but municipal inspections increased the formalization rate by 22 to 27 percentage points. In Peru, Alcãzar et al. (2010) find that offering a subsidy for the cost of obtaining a municipal license led to 10 to 12 percent of informal businesses obtaining it. In Malawi, Campos et al. (2015) find that offering assistance with costless business registration and (separate) tax registration had large impacts on business registration, with 75 percent of those offered assistance obtaining a business registration certificate, but only a limited impact on tax registration.

Evidence on the effect of becoming formal on business performance suggests that whereas some firms may see some benefits from formalization, like being able to advertise more or issuing receipts, for most firms these benefits are smaller than the costs associated with formalization in terms of additional taxes. The study in Sri Lanka (De Mel et al. (2013)) is the only one to our knowledge that finds some impact of formalization on business profits, and this impact is driven by a handful of firms for which profit increased significantly.

Bank financing is often hypothesized as one of the major channels through which formalization can improve firm performance, yet randomized experiments measuring formalization in Sri Lanka (De Mel et al. (2013)) and Brazil (Andrade et al. (2013)) have recorded no significant increase in the number of bank loans awarded to firms becoming formal. One possible reason is that newly formalized firms require more than just their formal certification, but also facilitation and access to banks. In Malawi, one version of the program evaluated in Campos et al. (2015) offered some support to open a bank account as an additional incentive to formalization. Short-term effects showed some impacts on the number of businesses owning a bank account, on savings, and on the use of complementary financial products.

In April 2014, the Government of Benin launched the pilot phase of the *entreprenant* status, a simplified and free legal regime offered to small informal businesses to enter the formal economy. This paper presents the short term results of a randomized impact evaluation testing three different versions of the *entreprenant* status on business registration decisions, each version including specific incentives to registration.

The first version of the program tests whether simplifying entry regulation and providing information alone could be effective in Benin, where formalization also means tax registration.<sup>2</sup> Specifically, in-person visits were conducted with informal businesses to present the new *entreprenant* status available freely and in a day. The second version of the program is based on the idea that businesses may need additional incentives to take up formalization and, once formalized, additional support to access benefits and grow. In this version of the program, businesses that chose to formalize had the option to benefit from business training, counseling services, and some support to open a business bank account with advantageous conditions. Finally, in the third version of the program, businesses received tax mediation services as an additional incentive to formalization. The premise was to provide some form of protection in case the tax inspectors abused their power with newly formalized businesses. This last version of the programs aims to test the extent to which the fear of the tax administration is a barrier to formality. To our knowledge, this is the first attempt to answer this question experimentally.

For the study, 3,600 informal businesses operating with a fixed location in Cotonou, including Dantokpa market (one of the largest markets in West Africa), were randomly allocated between these three treatment groups and one control group. In comparison to the studies mentioned above, this sample is composed of micro and small businesses with 1.2 employees on average and a monthly profit of CFAF 47,000 (US\$213).<sup>3</sup> One year after the program started, all 2,400 informal businesses selected in the treatment groups received at least one visit from an advisor who presented the program and the potential benefits associated with formalization (specified by group). The advisors were affiliated with the CGA in Cotonou (*Centres de Gestion Agréés*), a semi-public organization that specialized in business trainings. About 14 percent of businesses in groups 2 and 3 have started to use business training or counseling, and only 3% opened a bank account.<sup>4</sup>

In this paper, we focus on the program's impact on formalization one year after the program rollout. Important components of the program, like the business trainings, the counseling and the bank services were implemented more than 6 months after the program started. The impact that formalization could have on other outcomes like business performances or access to

 $<sup>^{2}</sup>$  A majority of informal businesses also pay taxes but usually the amounts to be paid are much smaller for informal firms. See Section 2 for a description of tax implications of formality.

 $<sup>^{3}</sup>$  USD 1 = XOF 218.82 in PPP terms in 2011-2015 (World Bank).

<sup>&</sup>lt;sup>4</sup> Results on bank account opening are based on data from Bank of Africa (updated to November 2014) and Orabank (updated to April 2015).

banking may therefore take more time to materialize. Data from a second follow-up survey expected in March 2016 will explore the impacts on these other outcomes.

One year after the program launch, all versions of the program had a significant impact on formalization rates. The impact was 9.1 percentage points in the first treatment group providing only registration simplification and information; 13 percentage points in the second group, which also provided business training, counseling and bank support services; and 15.8 percentage points in the last group, which also added to the previous group's benefits the provision of tax mediation services. Differences between groups are all significant, suggesting that businesses valued the additional incentives provided. These impacts are slightly higher than those measured in studies conducted in other contexts, especially given the fact that formalization is linked to tax registration in Benin. The formalization rate in the control group was less than 1 percent, suggesting that in the absence of the program, this type of informal businesses would not usually formalize.

The program had a higher impact on male businesses owners, those with more education, operating outside Dantokpa Market, in sectors other than trade, and that before being offered the incentives to formalization had characteristics similar to businesses that were already formal. Program impacts in all these sub-populations were significantly higher than in the rest of the sample, but formalization rates still did not exceed 23 percent. As a result, the majority of the informal sector is likely to remain informal, even after this policy change.

Two factors may explain why we did not observe greater impacts. First, the lack of proper identification was a barrier: only 53% of business owners had a passport or an ID card, the official documents required to formalize. Second, despite the impact on formalization, the program did not impact much the perception of formalization for many firms. For example, significant shares of beneficiaries still considered registration time and cost excessively high, and procedures complicated.

The remaining sections of the paper are organized as follow: Section 2 presents the potential benefits and costs associated with formality in Benin, section 3 details the *entreprenant* program, the sampling, the study design and the data. Section 4 describes program implementation during the first year and take-up on program components. Section 5 presents

program impact on formalization, section 6 analyzes mechanisms that could explain the results on formalization and section 7 concludes and presents the next steps.

## 2. Potential benefits and costs of formalization in Benin

Following technical assistance provided by the World Bank Group (WBG) and other donors, OHADA member countries (*Organisation pour l'Harmonisation en Afrique du Droit des Affaires*) adopted a revised General Commercial Law in December 2010, which came into effect in May 2011. The new law, immediately applicable to all OHADA members, introduced the *entreprenant* status, a simplified legal regime specifically designed for small entrepreneurs, whose intended objective is to facilitate the migration of businesses operating in the informal sector into the formal sector. However, the law did not make explicit how the *entreprenant* status practically functioned, nor the specific combination of incentives that it would include, instead allowing each country to fill in the vacuum through *ad-hoc* secondary legislation and institutional changes. Benin, as a member of OHADA, is the first OHADA country to roll out a pilot version of the *entreprenant* legal status.

The *entreprenant* status can apply to a physical person running a micro or small business involved in any type of activity. Formalization with this new status is easy, free of charge and takes only one business day. The introduction of the *entreprenant* status is part of a broader effort from the Government of Benin to simplify and reduce the costs of formalization. Reforms of other existing legal status were implemented a few months before the creation of *entreprenant* status, and included the creation of a one-stop shop for formalization, and a significant reduction of registration costs associated to the main existing legal status. The registration cost for individual enterprises dropped from CFAF 65,000 (US\$297) to CFAF 10,000 (US\$46) and from CFAF 225,000 (US\$1,028) to CFAF 17,000 (US\$78) for limited liability companies (only the *entreprenant* status is totally free of charge). For all statuses the time to register was reduced to one business day. As these reforms (including the creation of the *entreprenant* status) were implemented recently, information on the new conditions to formalize was not likely to be known by the majority of informal businesses operating in Cotonou at the time of the start of the program.

Formalizing in Benin means to choose a legal status and register at the chamber of commerce (GUFE, *Guichet Unique de Formalisation des Entreprises*). It offers some potential benefits (presented in Table 1) depending on the type of status chosen. Most of these potential benefits are related to the possibility to apply for bank services, or to access to new markets like government and large companies' contracts. The *entreprenant* status gives access to all advantages except the rights to export and to access large public contracts. It is explicitly targeted to micro and small businesses managing one type of activity with a limited turnover.<sup>5</sup> Businesses with multiple activities or with turnover greater than a threshold in two consecutive years will lose *entreprenant* status and adopt the *individual entreprise* status.

When they formalize, businesses get a unique fiscal identifier and are registered with the tax administration. Accordingly, the main potential cost of formalization is related to taxes. In Benin, the link between formalization and taxes is complex and varies according to the business. In theory, all businesses with a fixed location would pay taxes even if they were informal. Before the reform of the tax system affecting microentrepreneurs was introduced in 2015,<sup>6</sup> which will be used to calculate the tax owed in 2016, there were four different tax regimes that could apply to informal businesses in Cotonou, depending on their location and economic activity.<sup>7</sup> The regime most commonly applicable was the TPU (*"Taxe Professionnelle Unique"*) and was calculated based on the rental value of the business premises. However, in the majority of cases taxpayers do not have a lease contract, the only official and opposable proof of rental value. As a result, the law assigned the tax administration as responsible for assessing the rental value. This assessment often left a door open for discretion. In practice, tax inspectors estimated businesses' ability to pay based on their appearance and on discussion with business owners.

<sup>&</sup>lt;sup>5</sup> The OHADA General Commercial Law defines the *entreprenant* as having an annual turnover below XOF 30 million (approximately USD 137,100) for trading activities, XOF 20 million (approximately USD 91,400) for crafting activities (artisans), and XOF 10 million (approximately USD 45,700) for services. Once the small business adopts the *entreprenant* status, the turnover threshold should not be exceeded for more than two consecutive years.

<sup>&</sup>lt;sup>6</sup> On December 2014, the Beninese Parliament adopted a new MSE tax regime. This regime introduced the Synthetic Professional Tax (TPS: *Taxe Professionnelle Synthètique*) which replaces the four taxes that micro and small enterprises were subject to before the reform. This new tax introduces a major shift by changing the basis of tax calculation from the rental value to the use of turnover. This reform creates more predictability and transparency in the calculation of the amount of tax due and prevents small businesses from abuses of tax officers. MSEs will start paying the TPS in 2016 based on their 2015 turnover. All *entreprenants* will pay the TPS.

<sup>&</sup>lt;sup>7</sup> The four tax regimes were the following: "*Taxe Professionnelle Unique*" (TPU), "*Taxe Unique sur les Transports Routiers*" (TUTR), "*Régime du forfait des revendeurs de tissus et divers*", and "*Régime du bénéfice réel simplifié*".

Data collected before program implementation, presented in Table 2 (see section 3.2 below for a description of the database), suggest that taxes were more important for formal businesses for two reasons: first informal businesses were less likely to pay taxes, and second when they were paying taxes, they represented a lower share of their profit.<sup>8</sup> Slightly more than half (55 percent) of the informal businesses targeted were paying some form of tax in 2013. For formal businesses, it was the great majority (84 percent). One explanation for this difference is that informal businesses can possibly avoid paying taxes if their businesses are closed when tax inspectors visit the area. For formal businesses, it is more difficult because the tax administration knows that their businesses in 2013 represented 13 percent of annual profits for formal businesses, and only 5 percent for the informal ones. If we only consider businesses which paid some taxes in 2013, the amounts paid represented 17 percent of annual profits for formal businesses and 9 percent for informal ones.

However, the main objective of the Government of Benin with the *entreprenant* program is not to increase the tax collected in the short term. When they formalize, businesses can benefit from tax exemptions under certain conditions. Businesses which also register to the CGA (an association providing business counselling and account certification) can benefit from a full tax exemption for the first year after formalizing, in addition to a reduction of 40% in the amount of taxes due for the following 3 years. As a result, the amount of taxes paid in the very short term may not be sizeable for many businesses.

## 3. Evaluation design

## 3.1. Program description

The *entreprenant* status can be considered as a package of incentives to formalization that the government offers to small business. The implementation of the *entreprenant* status in Cotonou offers the ideal conditions to study the impact of different packages of incentives on formalization decisions, the legal environment in which firms operate, and how businesses interact with public authorities. Also, it shows the impact of different business and entrepreneur

<sup>&</sup>lt;sup>8</sup> It is difficult to compare informal and formal businesses since levels of profits are very different. Indeed, the difference in tax rates may only reflect progressivity with profits. However, when we compare tax rates at similar levels of profit, the correlation between tax rate and formality is strongly negative.

characteristics as determinants of formalization, and whether specific incentives carry any tangible benefits for the firm and its growth prospects.

The impact evaluation of the *entreprenant* status is a randomized control trial that tests the three following packages of formalization incentives.

Package A of incentives includes the following components:

- Provision of information on the new registration system, given in-person to business owners;
- Provision of tax information and clarification on the existing tax regimes applicable to the *entreprenant*.

*Package B* of incentives includes the following components:

• Services and training: services include support to entrepreneurs in the formalization process and, for example, help in drafting financial statements, business plans, and bookkeeping; training include basic accounting and business management, and initiation to legal and tax obligations;

• Bank services: creation of a bank account (checking or saving) with a commercial bank. *Package C* of incentives includes the following components:

• Provision of tax mediation services: support on preparing the tax declarations; provision of safeguards against arbitrary practices from the tax administration through mediation services between *entreprenants* and the tax authority.

The three different packages of incentives were delivered by the following institutions:

- GUFE, the one stop-shop for business registration based in Cotonou. Two additional staffs were hired to assist with *entreprenants* registration;
- CGA, formed by two non-profit organizations based in Cotonou whose mission is to provide small and medium enterprises with business management, accounting, and tax consulting services. Twenty-four advisors, one hotline assistant, and one supervisor were hired at CGA to implement the program;
- Two commercial banks, Bank of Africa (BoA) and Orabank.

The following paragraphs describe in details all program components.

# 3.1.1. Package A – Information on the *entreprenant* status

CGA advisors visited each business selected in treatment groups and explained the benefits of becoming an *entreprenant*. Different leaflets were given and explained to business owners: one leaflet described the *entreprenant* status, its advantages and requirements, one leaflet explained the registration process at GUFE, and one leaflet explained the different tax regimes applicable to *entreprenants* and how to calculate taxes due within each regime (see section 2).

The informal businesses that decided to formalize submitted an application at GUFE to obtain the *entreprenant* card. When necessary, CGA advisors helped *entreprenants* with the formalization process at GUFE, including filling in the declarations and preparing all the required accompanying documents.

#### **3.1.2.** Package B – Provision of business services and trainings

Following the first visit to each business, CGA advisors organized a second visit to deliver a 1-2 hour personalized training session. All businesses in groups 2 and 3 were eligible for this second visit even if they had not yet formalized. In addition to the business training, the purpose of this visit was to nudge business owners to participate to the *entreprenant* program. Once the informal business registered at GUFE and received the *entreprenant* card, it agreed with the CGA advisors on additional specific trainings that it might need. CGA advisors delivered the business trainings and services via a dedicated "*entreprenant* unit," using modules customized and developed with WBG support.

The advisory services and trainings were carried out on a one-on-one basis with the businesses at their premises and during group sessions organized at CGA. The group sessions included four workshops: three mandatory and one optional. The mandatory workshops were: (a) basic accounting, (b) initiation to tax obligations, and (c) financial education. For the optional workshop, businesses were invited to choose between (i) basics of microenterprise management, (ii) initiation to sales development and access to markets, and (iii) basic of business plan development. Each workshop lasted three consecutive half-day. Once the business owner completed the fours workshops with the CGA, he/she received an official diploma, and a sticker acknowledging that he/she received the training.

#### **3.1.3.** Package B – Support with bank services

In addition to the information on the *entreprenant* status and the training and counselling services, businesses in groups 2 and 3 also received support from CGA in opening a bank

account. This support was not mandatory, and the *entreprenant* could decide to open a bank account autonomously. The bank account is directly linked to the *entreprenant* activity and it is not a personal account.

The bank partners of the impact evaluation (Orabank and Bank of Africa) designed a specific banking product for the *entreprenant*, with dedicated services and simplified banking access conditions, including debit card, bank account consultation with mobile phone, cash transfers, SMS-banking, internet banking and mobile money. The *entreprenant* bank accounts in both banks are cheaper than what businesses can usually get (around CFAF 1,000 per month (US\$4.6) against CFAF 2,000 (US\$9.1) and do not require any initial deposit, whereas business bank accounts usually do in Benin. CGA advisors assisted the *entreprenant* in presenting the information needed to open a bank account and give the necessary training to the entrepreneur on how to use the account.

# 3.1.4. Package C – Provision of tax preparation support and tax mediation services

For businesses in group 3, the CGA tax advisors offered as an additional incentive to formalization, tax preparation support and protection in case tax inspectors abuse their power when assessing and collecting the taxes owed.

Those businesses that formalized under the third group, were offered help in preparing the tax forms (including tax returns and supporting documentation). However, given that most businesses were subject to the TPU, and that the amount of TPU to be paid by a given business is determined by the tax administration without any form being filled by the business, this "offer" was not technically implemented. It will, however, be implemented starting in 2016, when the new tax regime comes into force. The advisors also left their contact information in case the *entreprenant* had any complaints about future tax payments and inspections.

Advisors then took action when an *entreprenant* asked for help in interacting with the tax administration, and they served as intermediary between the tax administration and the *entreprenant* in order to provide solutions that satisfy both parties, in case of complaints.

#### **3.2.** Sample selection and study population characteristics

In order to select the study population, a listing survey was conducted in Cotonou in March-April 2014. This survey was designed in order to obtain a representative sample of all businesses operating in Cotonou including Dantokpa market. All businesses with fixed location, excepting international and nationwide companies, were targeted.

Sampling protocols for inside and outside the market were different:

- *Sampling framework for Danktopa market:* we used a precise map of the market made by SOGEMA (*Société de Gestion des Marchés Autonomes*), the public company managing markets in Benin. This map allowed to divide geographically the market in small areas. We have then randomly selected areas in the markets in which 50% of the businesses (with fixed location) where sampled for the survey.<sup>9</sup>
- Sampling framework for other neighborhoods of Cotonou: We were able to obtain detailed maps of each of the 144 neighborhoods in Cotonou. Those maps allowed identifying the easy identification of *ilots* (blocks), the official administrative unit within a neighborhood. We used this administrative unit as a reference for the listing survey sampling. We then used information given by the tax administration (and confirmed by the survey company) in order to characterize neighborhoods as high or low firm density areas. We randomly sampled 38% of *ilots* in high density neighborhoods and 10% of the *ilots* in low density neighborhoods. In each *ilots* 68% of businesses where sampled for the survey in average.

Overall, 19,246 businesses were listed by surveyors. The listing survey allowed us to estimate the total number of businesses operating in Cotonou (with a fixed location, excluding nationwide businesses and liberal professions) to approximately 68,500, including around 5,000 in Dantokpa market.<sup>10</sup> Among those 19,246 businesses, 9,938 businesses were randomly selected to be surveyed. 7,945 (80%) businesses were successfully surveyed, 1,000 (10%) businesses refused to be surveyed, and 995 (10%) businesses were dropped because the business owner was not available or not reached after 4 attempts. Figure 1 details the listing survey results inside and outside the market.

<sup>&</sup>lt;sup>9</sup> Few areas were excluded from the sampling frame because they almost exclusively included businesses selling illegal products (i.e. taint oil, medicine, and voodoo products) or by large formal businesses.

<sup>&</sup>lt;sup>10</sup> As previously indicated, some sections of Dantokpa market were not included in the listing survey. Therefore, the total number of firms in Dantokpa is probably much higher.

From the 7,945 businesses surveyed, a population of 3,600 businesses was then selected to participate to the study based on the following goals:

- Drop businesses already formal
- Drop businesses that will probably not cooperate in the future or which will be probably difficult to find (i.e. businesses that refused to provide information on profits or turnover during baseline survey)
- Trim the database from (a) businesses very close to formalization who would have formalized anyway and (b) businesses very far from formalization which would not be interested by the program
- Remove businesses that ever got a loan from a commercial bank that will most probably not been interested by the program (less than 3% of informal businesses)
- Reduce the standard deviation of the main outcomes (profit and turnover)<sup>11</sup>
- Include a sufficient number of businesses in Dantokpa market.<sup>12</sup>

Table 2 shows descriptive statistics for businesses selected in the sample and businesses not selected.

The study sample is composed by relatively small businesses with 1.2 employees on average. The great majority of businesses (86%) were created over a year before being surveyed, 55% of businesses were involved in trade activities, 26% worked in services, and 16% were craftsmen. 63% of businesses sampled for the study were owned by women. This reflects the high share of female owners in Dantokpa market. Around 30% of businesses owners never went to school, and less than 20% of the businesses were keeping some type of accounting. As described previously, the majority of businesses paid some form of tax (55%), but the great majority complained about not knowing in advance the amount of taxes owed to the tax authority (74%).

In comparison to similar studies in other contexts, this study population is composed of businesses with slightly smaller sizes. In the study in Malawi (Campos et al, 2015), businesses

<sup>&</sup>lt;sup>11</sup> Outside Dantokpa market we excluded businesses with sales or profit lower than XOF 12,000 (USD 55), profit greater than XOF 150,000 (USD 685) or sales greater than XOF 400,000 (USD 1,828). In Dantokpa market, we excluded businesses with sales or profit lower than XOF 10,000 (USD 46), profit greater than XOF 200,000 (USD 914) or sales greater than XOF 500,000 (USD 2,285).

<sup>&</sup>lt;sup>12</sup> We choose to take 22% of total study population from Dantokpa market to have the same share of businesses from the market than in the 2008 firm census (INSEA, 2008).

had on average two employees and monthly profit of US\$214, while in the study in Sri Lanka (de Mel et al. (2013)), businesses had on average three employees and monthly profit of US\$300.

Businesses selected for the study have very similar characteristics to the whole population of informal businesses surveyed, and the overall study shows good external validity.

Formal businesses had on average 3.2 employees and monthly profits of around CFAF 210,000 (US\$960), while informal businesses had 1.1 employees and a monthly profit of CFAF 46,000 (US\$210). Formal businesses also had higher access to the banking system: 80% of them owned a bank account, whereas only 20% of informal businesses did.

#### **3.3. Evaluation Design**

The 3,600 informal businesses were randomly allocated into three treatment groups and one control group. The first group of informal businesses received package A of incentives, the second group packages A and B of incentives, and the third group packages A, B and C.

The randomization was done in the office using STATA. The following methodology was used for stratification:

- (1) 16 strata were created using the following variables: business owner gender, business operating in Dantokpa market, trader and business owner a bank account.
- (2) Inside each stratum a Z-score was created as the average of standardized profits, turnover and number of employees. Based on this Z-score, triplets of businesses were created and inside each triplet, businesses were randomly allocated to 3 groups: group 0, group "1 and 2", and group 3. Each group included 1,200 firms.
- (3) The 1,200 businesses in group "1 and 2" were randomly allocated into group 1 with 300 businesses and group 2 with 900 businesses.

As a result, 300 businesses were allocated to group 1, 900 to group 2, 1,200 to group 3 and 1,200 to the control group. Figure 2 describes the organizational chart of the interventions.

We decided to allocate fewer businesses in groups 1 and 2 than in group 3 and in the control group for statistical power consideration. We were interested in estimating both the impact of the different packages of incentives on formalization and in estimating the impact of formalization on business performances. For the second question, we needed the take-up rate on formalization to be sufficiently high to have enough statistical power. The literature on the impact of business regulation reforms suggests that programs which only simplify and cut the costs of formalization have very limited impact, and we therefore anticipated a small impact on formalization in group 1. We wanted to be able to check that in the context of Benin, but we included more businesses in treatment groups offering more incentives for which we expected a higher impact on formalization.

Table 3 presents the results of balance checks of baseline characteristics across the different treatment groups and control group. Overall, it shows that all groups are relatively well balanced with respect to observable characteristics: the number of tests that are statistically significant is close to what should be expected due to chance (2 out of 15 tests for the joint tests of all coefficients are equal to zero are significant).

# 3.4. Data

Three main sources of data are used for this study: administrative data on formalization and program implementation, in-person quantitative surveys with business owners, and qualitative data with study participants and implementing agencies.

Our main measure of formalization is based on monthly administrative data on business registration provided by the GUFE. This database includes the complete list of all newly registered businesses for all legal entities. We matched this list with the list of business owners who participated to the program. Two important features characterize this measure of formalization:

• Formalization rate is captured for all legal entities: Most businesses in the control group were not aware of the new *entreprenant* status, so they were probably more likely to choose another legal status if they wanted to formalize. For this reason we choose to define formalization as registered at the chamber of commerce and not to only focus on the *entreprenant* status.

Matching businesses in the GUFE database with businesses in the program database involved several steps. In order to match two businesses, we used the variables business owner names, business address, gender of the owner, sector of activity and phone number; however, in some cases name spellings were different in both databases, and most of the times addresses were not precise. For this reason, we used a conservative definition of a match between the business names in the two databases as "two businesses with a close surname, and at least one close first name, and either the same phone number, or the same sector of activity and an address in the same neighborhood." Using this definition, the likelihood that a business in the program database was considered as formal, whereas the business was in reality not formal, was very low (see Annex 1 for additional details of the matching process). The opposite case (i.e. a business was considered as informal, whereas it is in reality formal) is however more likely, so this measure of formalization most probably underestimates the actual number of businesses which formalized in all groups. This does not imply that the measure of program impact is biased, but just that formalization rates in both treatment and control groups are underestimated by the same percentage.

Other main outcomes on business performances (profits and turnover) and intermediate outcomes like business knowledge and practices, taxes and banking were measured through inperson interviews with business owners. The first survey of the selected sample of businesses was conducted in March-April 2014 prior to program implementation and was also used for the sampling. A follow-up survey was conducted in April-May 2015 to measure short-term effects one year after the program started. Attrition at this survey was 11.9 percent and was not correlated with treatment status. One year after the baseline survey, 6.1 percent of the businesses had closed their operations. Business closure is also not correlated with treatment status. Table 4 presents survey rate, closure rate and reason for attrition by groups.

Program implementation data were also collected to better understand the quality of services delivered. These included detailed monitoring data from CGA and qualitative surveys with implementing agencies and program participants. 30 semi-structured qualitative interviews were conducted with program participants at different stage of the program, a qualitative surveyor was also regularly sent with the CGA advisors to assess if the study design was respected (29 surveys). In addition, 61 qualitative interviews were conducted with business owners not selected for the program to monitor potential externalities of the program. Finally,

focus groups were conducted with the main implementing agencies (CGA, GUFE and both commercial banks).

A final in-person follow-up survey will be conducted in March 2016 to measure the mediumterm effects after two years from the beginning of the program, in particular the impact on business performance.

#### 4. Program implementation

Data from the implementing partner, in addition to quantitative and qualitative data, suggest that the program was implemented consistently with the study design. Treatment allocation was respected for all businesses, and all components of the program were effectively offered to almost all program beneficiaries.

The advisors from CGA delivered the program to each business owner following four main steps:

- (1) First visit: A CGA advisor conducted a first visit to each business to explain the benefits of becoming an *entreprenant*, specific by group, and to distribute informational leaflets. If a business owner was not present on the day of the visit, the CGA advisor attempted to call the owner on the phone. If the owner could not be reached, the CGA advisor made another attempt by trying a visit or a call in different moments of the day. After four attempts (visits or calls), the business was considered as not interested.
- (2) Second visit: For businesses in groups 2 and 3, the same CGA advisor called, arranged, and confirmed a meeting, which took place approximately two weeks after the first visit, and provided 1-2 hours of personalized training. If a business owner was not present on the planned day of the second visit or could not be reached, the CGA advisor made another attempt by trying a visit or call in different moments of the day. After 3 attempts (visits or calls), the business was considered not interested. Registration at GUFE was not mandatory to be eligible to this second visit. In group 3, the CGA advisor devoted additional time in reviewing the procedures to calculate the taxes to pay, and the option of receiving tax mediation help, if necessary.
- (3) Formalization decision: After having received the first and/or the second visit, business owners decided whether or not to register as *entreprenants* at GUFE.

(4) Provision of additional benefits: Businesses in groups 2 and 3 could also register to CGA, and receive counselling and business training (group sessions). Businesses in group 3 could benefit from tax mediation services with CGA, if needed. Finally, businesses could open a bank account with specific conditions at BoA or Orabank.

The program was implemented on a rolling basis: CGA advisors started to reach out to informal businesses in April 2014, and completed both visits in February 2015. The advisors are currently offering group trainings and counselling services to the *entreprenants* that requested it after the registration.

Between April 2014 and January 2015, 2,400 "first visits" (100% of total) were attempted by CGA. The take up rate for the first visit was remarkably high and 2,344 were completed with success (98% of total). First visits were considered as not completed successfully when CGA advisers were not able to locate the business. Between April 2014 and February 2015, all businesses who received a first visit in groups 2 and 3 were offered a second visit by CGA. Only 932 of these second visits were completed with success (44% of total). The main reasons for this relatively low take-up rate were that many businesses were not interested by the second visit, or did not have time to receive it. Most of the group training sessions applicable to businesses in groups 2 and 3 were conducted after September 2014. The time lapse between the first and the second visits was much greater than originally planned (3 months in average instead of 2 weeks) due to logistical constraints, and because it often required several trips to the business to complete the second visit successfully.

As of April 2015, 251 businesses registered with CGA (12% of total in groups 2 and 3), and 143 businesses participated in a group training session at CGA (7% of total in groups 2 and 3). Since businesses had first to register with GUFE in order to be eligible to register at the CGA, and thus receive the trainings, the percentage of eligible businesses that did register with CGA is sizeable. In fact, 78% of the businesses in groups 2 and 3 that formalized (322 businesses in total) decided to register with the CGA, and slightly less than 45% decided to obtain trainings. Business owners in groups 2 and 3 who decided to register as *entreprenant* had also the possibility to open a bank account at BoA or Orabank. As of April 2015, 69 businesses opened

a bank account (3.2% of total).<sup>13</sup> Panel A of Table 5 summarizes the program implementation progress until May 2015.

Qualitative information collected during program implementation<sup>14</sup> with beneficiaries suggests that the program was implemented following the study protocol and in particular that the formalization process with the *entreprenant* status was considered as simple and fast and was free. Panel B of Table 5 shows quantitative data from follow up survey and confirms that the formalization process was fast and cheap for businesses in treatment groups. More than 85 percent of businesses that benefited from the program declared that they did not pay anything to formalize (those who paid something in the treatment groups formalized with another status than the *entreprenant* status).

Qualitative work conducted few days or weeks after the businesses received a visit from the CGA suggests that the program understanding was relatively good, given the complexity of the intervention. However, data from the midline survey suggest that after some time, most businesses had forgotten the program. Only 36 percent of businesses in groups 2 and 3, and 24 percent of those in group 1, remembered the *entreprenant* program. Moreover, only 21 percent in groups 2 and 3, and 11 percent in group 1, were able to describe correctly what it is.

In the control group, only 6 percent of the businesses declared that they had heard about the *entreprenant* program, and 2 percent were able to describe it correctly. It suggests that only marginal externalities were generated by the program on those not directly targeted. This is consistent with qualitative interviews conducted with informal businesses not targeted by the program.

For businesses in group 3 that formalized, tax mediation services were implemented by CGA. Some *entreprenants* reported that the tax administration requested for tax payments higher than expected. The CGA advisors helped them to solve these issues as they arose. Only 2 cases happened in the first year of program implementation.

<sup>&</sup>lt;sup>13</sup> For Orabank, data are available until April 2015, and for BoA until November 30<sup>th</sup>, 2014.

<sup>&</sup>lt;sup>14</sup> 22 semi-structured qualitative interview with program participants after they received the first of second visits; 29 field visit with the CGA advisors during the first or second visits; 8 interviews with business owners who decided to formalize; 61 interviews with informal businesses not selected for the program; focus groups with all implementing agencies.

# 5. Results

#### 5.1. Estimation specification

Our estimation is at the business level, and involves the following specification for business *i*:

$$Y_{i,t=1} = \beta_0 + \beta_1 T 1_i + \beta_2 T 2_i + \beta_3 T 3_i + X'_{k,i} + \varepsilon_{i,t=1}$$
(1)

Where  $Y_{i,t=1}$  is the outcome variable (formalization),  $T1_i$  is an indicator for being assigned to treatment group 1,  $T2_i$  an indicator for being assigned to treatment group 2 and  $T3_i$  an indicator for being assigned to treatment group 3.  $X_k$  is a vector of strata dummy variables (one dummy variable for each triplet of businesses) (Bruhn and McKenzie, 2009) and  $\varepsilon_{i,t}$  is the error term.  $\beta_1$ ,  $\beta_2$  and  $\beta_3$  provide the intent-to-treat effect of being assigned to group 1, 2 and 3, respectively. This is the effect of being a business assigned to treatment 1, 2 or 3 relative to being a business in the control group.

#### 5.2. Impact on formalization

As detailed in section 3.4, our main measure of formalization correspond to the registration of the business with the chamber of commerce at GUFE (i.e. the registration was found in GUFE data). We think that this definition of formalization is preferable over others that use midline survey data because administrative data included information on all study population, whereas survey data only have information on those who were surveyed. Moreover, survey data are subject to declaration bias. However, the correlation between survey data and administrative data was high (0.64), and we show similar results using the survey data as well.

Table 6 presents the results on formalization one year after the program started. The Impact of the program on the formalization rate was 9.1 percentage points in group 1, 13 percentage points in group 2, and 15.8 percentage points in group 3. All these effects are statistically significant at one percent level. The effects in groups 2 and 3 are significantly higher than in group 1, and the effect in group 3 is significantly higher than in group 2: both sets of additional incentives included in package B (counselling, trainings and bank services) and in package C (tax mediation) seemed to be valued by informal businesses as incentives to register.

The formalization rate in the control group was only 0.9 percent. Therefore, in the absence of the program, only few businesses would have formalized. Alternative measures of formalization that combine survey and administrative data show consistent results on

businesses surveyed during the follow-up survey. Impact rates in groups 2 and 3 are always significantly higher than in group 1, and impact rates in group 3 slightly higher than in group 2 (but not significantly).

Table 7 shows heterogeneous impact by business characteristics. We find that male business owners formalized much more than female business owner. Slightly above 10 percent of businesses owned by women formalized in groups 2 and 3, compared to around 20 percent for those owned by men. This result could be correlated with the fact that a large majority of businesses operating in Dantokpa market are owned by women. However, column 7 of the table shows that it is also true outside Dantokpa Market for women not operating in trade.

In all groups, formalization rates were 7-9 percentage points higher outside Dantokpa market than inside the market. One explanation to this result is that formalization could be less attractive in the market as businesses are already registered with the public company in charge of all markets in Cotonou (SOGEMA- "Société de Gestion des Marchés Autonomes"). They are also more interconnected as they usually have representatives they can address in case of problems with the administration.

Businesses operating in the trade sector had lower formalization rates than in other sectors. This result is probably correlated with the fact that almost all businesses in Dantokpa market are traders, but it is also true outside the market.<sup>15</sup> One possible explanation which was mentioned during qualitative interviews is that before the program implementation, traders already had access to a "trader card" that provides a formal status with specific benefits (see Table 1), whereas no such specific card existed for other sectors.

The program was more effective on businesses with an owner who went at least to secondary school. This is not surprising if we consider that business owners with more education may benefit more from trainings and counselling services. Formalization also requires doing some basic accounting, which could be more difficult with a low level of education.

Formalization rates were higher among informal businesses that were similar to formal businesses before program implementation. Using species classification techniques (de Mel et

<sup>&</sup>lt;sup>15</sup> Results not shown but available upon request

al, 2010) we classified 18 percent of the businesses in the sample as "looking more like formal businesses before the program".<sup>16</sup> This result is especially large for group 3, as businesses with characteristics similar to formal businesses in this group formalized 20 percentage points more than the same businesses in the control group.

Finally, businesses that received more than one visit from a tax inspector in the year prior to program implementation were more likely to formalize. This result, which is only significant for group 3, may suggest that the program was perceived as a way to limit tax harassment.

## 6. What prevents more firms from formalizing?

We have seen that in the third treatment group, which combined all packages of incentives and in which the impact was the greatest, formalization rate was about 17 percent. This impact is greater than for similar programs in other contexts (Bruhn and McKenzie, 2014) in which formalization is also linked to taxes. But it means that even though this type of program had a significant impact, the majority of the informal firms still remain informal. This is true even if we consider specific sub-population like businesses operating outside the market or with an owner who went to secondary school, for which impact on formalization remained below 23 percentage points.

We think that some important factors could explain this result in our particular context:

- (1) The presence of other legal barriers to formalization;
- (2) The lack of success in changing the perception of formalization costs and benefits.

(1) Other legal barriers, like the lack of identification documents, may play an important role in explaining the results, as qualitative data from program beneficiaries and implementing agencies suggest. Moreover, data presented in Table 8 show that only 53 percent of business

<sup>&</sup>lt;sup>16</sup> Looking like a formal business owner is based on the predicted probability of being formal from a logit of formality status on baseline characteristics. This logit uses the data collected during the listing/baseline survey on 7,829 businesses who accepted the survey. Among them, 608 (7.8%) were formal at the time of the survey. We used the following baseline characteristics in the logit: operating in TOKPA market, gender, age, only primary education, only JHS or SHS level, higher level of education, operating in services, craftsman, business created less than 1 year ago, firm connected to electricity network, total number of employees, firm is doing some accounting, have done any advertising in the last 6 months, log of total amount of sales in an average week, log amount of last month profit, firm owner owns a bank account, the firm pays taxes, have done any advertising in the last 6 months (and controls for missing levels of these variables). Using the "predict" command in STATA, we end up classifying as "looking more like formal" 654 (18.2%) businesses out of the 3,600 in the study sample.

owners have one of the legal ID required in the formalization process. The administrative process necessary to receive an ID card or a passport may have discouraged business owners to formalize. For other types of legal documents, like the birth certificate or the electoral card, ownership rates are much higher. Authorizing these other legal document for the formalization process could help releasing this constraint.

(2) Table 8 shows the perception that business owners have of the formalization costs and benefits. Three main types of cost were cited by respondents: increase in taxes (30 percent), high registration costs (23 percent), and complicated or time consuming registration process (23 percent). In the control group, 24 percent of the respondents answered that they do not know what costs are associated with formalization, and 20 percent do not think that there are any cost. Perceived benefits of formalization can be divided into four categories: access to new markets (35 percent), better reputation or social acceptance for the business (31 percent), increased access to bank services (24 percent), and reduction of the risk of been fined or asked for bribes (15 percent). 22.5 percent of the control group did not know what benefits are associated to formalization and 14 percent did not think that there were any benefits. The program did not have many effects on the perceived costs and benefits of formalization. We see that slightly less respondents in groups 2 and 3 considered the registration costs as a disadvantage (only significant for group 3), and that the program had a five percentage points impact on the share of respondents who answered that there are no costs associated to formalization. We also see a significant impact in groups 2 and 3 on the share of respondents who answered that formalization can give access to bank services.

This result is surprising given that, as detailed in section 4, the take-up rate of the first visit of CGA advisors to businesses was close to full coverage. Almost all business owners in treatment groups received at least one visit from a CGA advisor who detailed the benefits and costs associated to formalization and gave them program leaflets. Less than one year after that visit, most of them did not change their perception of formalization. Similar findings occurred in Andrade et al.'s (2013) experiment in Brazil, where they find few changes in knowledge about formalization after a communication treatment. Possible explanations could be that they did not trust the CGA advisors, that the message was too complicated, or perhaps that the question of formalization is not of first importance for this type of small businesses. The fact that one year after the program launch, only 30 percent of businesses in groups 2 and 3 and only 20

percent of those in group 1 remembered the *entreprenant* program could be an argument for this last point (see section 4 and Table 5).

# 7. Conclusions and next steps

These early findings show that the launch of the entreprenant status and assistance registering under it has resulted in an increase in formalization, with this increase greater when the status is coupled with access to business services and tax mediation. A key question is then whether it is worth it for these firms to formalize from either a firm viewpoint or a public policy viewpoint. This depends on the extent to which formalization enables firms to access finance, gain more customers, and grow, and the extent to which incorporating firms into the formal sector results in increased tax payments or other public policy benefits. Our ongoing impact evaluation will attempt to measure these channels.

# References

Alcãzar, L., Andrade, R. and Jaramillo. M. (2010): "Panel/tracer study on the impact of business facilitation processes on enterprises and identification of priorities for future business enabling environment projects in Lima, Peru – Report 5: impact evaluation after the third round." Report to the International Finance Corporation. World Bank Group, Washington, DC.

Andrade, G. H., Bruhn, M., and McKenzie, D. (2013): "A helping hand or the long arm of the law" Experimental evidence on what governments can do to formalize firms", *World Bank Economic Review*, forthcoming.

Bruhn, M. (2013): "A Tale of Two Species: Revisiting the Effect of Registration Reform on Informal Business Owners in Mexico" *Journal of Development Economics* 103: 275–83

Bruhn, M., McKenzie, D. (2009): "In Pursuit of Balance: Randomization in Practice in Development Field Experiments", *American Economic Journal: Applied Economics*, 1(4): 200-32.

Bruhn, M., McKenzie, D. (2014): "Entry regulation and formalization of microenterprises in developing countries", *World Bank Research Observer*, 29(2): 186-201.

Campos, F., Goldstein, M., and McKenzie, D., (2015): "Short-term impacts of formalization assistance and a bank information session on business registration and access to finance in Malawi," *Policy Research Working Paper Series* 7183, The World Bank.

De Giorgi, G., and R. Rahman. (2013): "SME's Registration: Evidence from an RCT in Bangladesh." Stanford University, Stanford, CA, and World Bank, Washington, DC.

De Mel, S., D. McKenzie, and C. Woodruff. (2013): "The demand for, and consequences of, formalization among informal firms in Sri Lanka." *American Economic Journal: Applied Economics* 5 (2): 122–50.

De Mel, S., D. McKenzie, and C. Woodruff. (2010): "Who are the Microenterprise Owners?: Evidence from Sri Lanka on Tokman v. de Soto" in *International Differences in Entrepreneurship*, J. Lerner and A. Schoar (eds.), pp.63-87, 2010.

De Soto, H. (1989): The Other Path, New York: Harper and Row Publishers.

Farrell, D. (2004). "The Hidden Dangers of the Informal Economy." *McKinsey Quarterly* 2004, no. 3:26–37.

INSAE, (2009): "Comptes nationaux," INSAE, Cotonou.

La Porta, R., Shleifer, A. (2014): "Informality and Development", *Journal of Economic Perspectives* 28(3): 109-26.

La Porta, R. Shleifer, A. (2014): "The Unofficial Economy in Africa," NBER Chapters, in: African Successes: Government and Institutions National Bureau of Economic Research, Inc.

Levy, S. (2008): "Good Intentions, Bad Outcomes: Social Policy, Informality and Economic Growth in Mexico", *Washington: Brookings Institution Press* 

Maloney, W. (2004): "Informality Revisited", World Development 32(7): 1159-78

McKenzie, D. (2012): "Beyond Baseline and Follow-up: The Case for more T in Experiments", *Journal of Development Economics*, 99(2): 210-21.

Perry, G., Maloney, G. Arias, O. Fajnzylber, P. Mason, and A., Saavedra, J. (2007): "Informality: Exit and Exclusion". World Bank Latin America and Caribbean Studies: World Bank, Washington D.C.

World Bank (2015): Doing Business 2015: Going Beyond Efficiency.



# Figure 1: Sampling strategy and Survey completion rates





# Annex 1: Matching program data to administrative data on formalization

This Annex describes the protocol to match businesses in the administrative database on formalization provided by the GUFE (around 550 businesses every month) with the program data (3,600 informal businesses prior to the program start).

# Information available:

We had the following information in both databases:

- Surname of the business owner. It can be written with different spellings in each database.
- Between 1 and 5 first names. In GUFE data we usually have more than one first name. In the program data we only have one first name in most of the cases.
- The business activity as described by the owner (no codes). The business activity is missing for 30% of businesses in the GUFE data.
- Business addresses. In the GUFE data addresses were given by the business owner whereas in the program data, we are using "official" addresses used by the tax administration (there are 144 neighborhoods in Cotonou). In practice only neighborhoods can be matched. In GUFE data, there are few missing variables and some cases for which the neighborhood does not belong to the official list of neighborhoods.
- Gender of the business owner.
- Phone number of the business owner

# **Definition of a match:**

We consider it to be a match if: (i) the surname, at least one first name, the activity, and the neighborhood match or (ii) if surname and at least one first name match and either the activity, the neighborhood or the phone number also match, and the others are missing (or does not exist for the neighborhood or the phone number).

# Method of matching:

We used first the Stata command "reclinck" designed for fuzzy name matching. This command uses record linkage methods to generate matching scores. For this first step, we used tree variables: surname, first name and gender. As a second step, we looked manually (in an Excel file) to all matches and validate each match only if names, activity and neighborhood were consistent in both databases.

The "reclink" command allows inputting different weights to match on each of the three variables used (surname, first name and gender). In order not to rely on the weights used, we reiterate the process with different weights until no additional matches were found.

Since it is possible that the first name in the program data corresponds to the second name in the GUFE data, we also reiterate the whole process for all combinations of first to fifth names. Surname and first name were inverted in one of the two databases. So we also reiterate the process with other combination of surname and first names.

# Checking that the matching method is working:

To assess whether our matching method is working efficiently, we used the following methods:

- First we looked at whether we could find additional matches using a more usual method of matching. That is looking at the two lists (sorted by surnames) and trying to find each business of the GUFE data in the program database. So it means looking mainly at businesses with surnames starting with the same letters. We were not able to find any additional matches.
- 2. Secondly, we looked at the proportion of business which formalized with the *entreprenant* status during the first 3 months after program launch. Indeed, most businesses which formalized with the *entreprenant* status should be also in the program data (in theory they are the only businesses aware of this new status). We matched 78% (119/153) of the newly registered *entreprenants*.
- 3. We then took the 34 businesses which formalized with the *entreprenant* status and were not matched with the program data and we tried harder to match these businesses. We took the program data and looked at all the businesses in the same neighborhood as the unmatched businesses. We were able to find 6 new matches. These 6 matches are very imperfect matches with surnames somewhat different and first name sometimes different. For two cases, the match was not done with our main method because the surname is missing in the program data.

Table 1 : Potentia	l Benefits and	<b>Costs of Form</b>	alization in Benin
--------------------	----------------	----------------------	--------------------

Type of status	Informal (fixed location)	Entreprenant status	Individual enterprises	Limited liability company	Trader card <sup>1</sup>
Cost of the status	n.a.	Free of charge	XOF 10,000 (USD 46)	XOF 17,000 (USD 78)	Extra XOF 5,000 (USD 23)
Time to register	n.a.	1 day	1 day	1 day	1 day
Maximum Turnover	n.a.	XOF 30 million for traders. XOF 20 million for craftsmen, XOF 10 million for services	No	No	No
Register more than one activity for the firm	n.a.	No	Yes	Yes	Yes
Needs to pay taxes	55% pay taxes	Yes with tax exemptions after formalization <sup>2</sup>	Yes with tax exemptions after formalization <sup>2</sup>	Yes with tax exemptions after formalization <sup>2</sup>	Yes with tax exemptions after formalization <sup>2</sup>
Open business bank account	No	Yes	Yes	Yes	Yes
Apply to a bank loan	Requires collateral	Yes	Yes	Yes	Yes
Export license	No	No	No	Yes (need to get an export card)	Yes (need to get an export card)
Can work with large private companies	Possible but complicate	Yes	Yes	Yes	Yes
Access to large public contract	No	No	No	Yes	Yes
Access to small public contract	No	Yes	Yes	Yes	Yes
Registered at the chamber of Commerce	No	Yes	Yes	Yes	Yes
Provide invoices to customers for tax purposes	No	Yes	Yes	Yes	Yes
Register more than one activity for the firm	n.a.	No	Yes	Yes	Yes

<sup>1</sup>: For the trader card, businesses also need to get the individual enterprise status. <sup>2</sup>: Businesses that formalize, registered with CGA, and that had not paid taxes before, have a tax exemption for the first year after formalization, in addition to a reduction of 40% in the amount of taxes due for the following 3 years.

# Table 2: Descriptive statistics on study population

	(1)		(2)		(3)		
	SELECTED Sa Mean [SI	mple )]	All informal businesses Mean [SD]		Formal busin Mean [S	nesses D]	
– Firm owner characteristics							
Female owner	0.629	3600	0.632	7093	0.419	608	
	[0.483]		[0.482]		[0.494]		
Age of the owner	39.454	3561	39.419 6959		43.637	589	
	[10.386]		[11.168]		[10.525]		
Firm owner has at least some formal	0.712	3595	0.708	7085	0.884	606	
education	[0.453]		[0.455]		[0.32]		
Firm characteristics							
Trade	0.549	3600	0.518	7094	0.584	608	
	[0.498]		[0.5]		[0.493]		
Services	0.262	3600	0.277	7094	0.26	608	
	[0.44]		[0.447]		[0.439]		
Craft	0.16	3600	0.17	7094	0.09	608	
	[0.367]		[0.376]		[0.287]		
Business created less than one year ago	0.144	3600	0.183	7094	0.137	606	
	[0.351]		[0.386]		[0.344]		
Firm area in m <sup>2</sup>	18.72	3594	18.322	7082	52.479	606	
	[43.479]		[50.788]		[106.493]		
Number of employee	1.176	3600	1.03	7094	2.961	608	
	[1.687]		[1.603]		[4.59]		
The firm does any form of accounting							
	0.179	3598	0.156	7093	0.642	604	
	[0.383]		[0.363]		[0.48]		
Amount of sales in an average week	60553.35	3600	82612.66	6643	542166.8	528	
(FCFA)	[56492.2]		[298607.1]		[4434990]		
Amount of profit in the last month	46671.9	3600	46419.43	6362	223041.2	490	
(FCFA)	[46558.47]		[141379.5]		[726068.4]		
Firm owner owns a bank account	0.222	3517	0.194	6931	0.789	582	
	[0.416]		[0.396]		[0.409]		
Firm pays taxes	0.547	3564	0.466	7009	0.836	597	
	[0.498]		[0.499]		[0.371]		
Amount of taxes paid in the previous	18742.54	3486	16655.86	6831	316636.4	533	
year (FCFA)	[27265.45]		[30725.88]		[2591065]		
Thinks that it's difficult or very difficult to	0.744	2669	0.765	4925	0.725	520	
know in advance now much taxes he/she	[0.436]		[0.424]		[0.447]		
Ratio tax/ annual profit for all businesses	0.051	3486	0.072	6178	0 128	445	
	[0.089]	0.00	[0,174]	01/0	[0.221]		
Ratio tax/ annual profit for businesses	0.094	1873	0 165	2862	0 169	372	
paying taxes	[0.104]	10/5	[0.286]	2002	[0.313]	572	

Notes: sources: listing-baseline survey March 2014

#### Table 3: Balance checks on baseline characteristics

	Difference between []							P-values of joint	
	Mean [SD]	and Control group		_	P-valu	e for differ	ence	test	
	in Control				_	Group 1	Group 1	Group 2	group1=group2=
	Group	Group 1	Group 2	Group 3	Ν	and 2	and 3	and 3	group3=0
Firm owner characteristics									
Female owner	0.629	0.001	0	-0.002	3600	0.717	0.363	0.414	0.63
	[0.483]	(0.003)	(0.002)	(0.002)					
Age of the owner	39.2	1.1	0.6	-0.2	3561	0.49	0.069*	0.091*	0.11
	[10.7]	(0.7)	(0.5)	(0.4)					
Firm owner has at least some	0.707	-0.034	-0.002	0.026	3595	0.374	0.063*	0.16	0.163
formal education	[0.456]	(0.032)	(0.02)	(0.018)					
Firm characteristics									
Trade	0.55	-0.007*	0	0	3600	0.103	0.071*	01	0.315
	[0.498]	(0.004)	(0.002)	(0.002)					
Services	0.259	0.007	-0.001	0.007	3600	0.761	01	0.586	0.937
	[0.438]	(0.023)	(0.014)	(0.013)					
Craft	0.166	-0.007	-0.01	-0.008	3600	0.929	0.97	0.921	0.884
	[0.372]	(0.023)	(0.014)	(0.013)					
Business created less than one year	0.137	-0.008	0.009	0.017	3600	0.527	0.323	0.645	0.596
ago	[0.344]	(0.025)	(0.016)	(0.014)					
Firm area in m <sup>2</sup>	19	6*	-0.2	-2.3	3594	0.076*	0.008***	0.282	0.055*
	[42.1]	(3.1)	(2)	(1.7)					
Number of employee	1.18	-0.044	-0.021	0.016	3600	0.826	0.519	0.52	0.86
	[1.681]	(0.092)	(0.057)	(0.051)					
The firm does any form of	0.173	0.013	-0.001	0.015	3598	0.631	0.917	0.306	0.678
accounting	[0.378]	(0.027)	(0.017)	(0.015)					
Amount of sales in an average week	60822	75	298	-1049	3600	0.929	0.618	0.337	0.751
(FCFA)	[57010]	(2255)	(1402)	(1253)					
Amount of profit in the last month	46195	142	309	1161	3600	0.944	0.633	0.521	0.788
(FCFA)	[44825]	(2134)	(1327)	(1186)					
Firm owner owns a bank account	0.22	0.004	0.004**	0.005***	3517	0.987	0.772	0.663	0.016**
	[0.414]	(0.003)	(0.002)	(0.002)					
Firm pays taxes	0.561	-0.014	-0.016	-0.022	3564	0.94	0.791	0.77	0.668
	[0.496]	(0.033)	(0.021)	(0.018)					
Amount of taxes paid in the	19451	-3883**	-281	-688	3486	0.097*	0.103	0.735	0.265
previous year (FCFA)	[28140]	(1953)	(1210)	(1076)					

Notes: sources: listing-baseline survey March 2014. Column 1: Standard deviations presented in brackets. Columns 2-4: coefficients and standard errors (in parentheses) from an OLS regression of the firm owner/firm characteristic on treatment dummies, controlling for strata dummies (dummies for each triplet). \*\*\*, \*\*, \* indicate statistical significance at 1, 5 and 10%.

#### Table 4: attrition and business closure at first follow-up survey

	Difference between []								P-values of joint
	Mean [SD]	and	l Control g	group		P-value for difference			test
	in Control					Group 1	Group 1	Group 2	group1=group2=
	Group	Group 1	Group 2	Group 3	Ν	and 2	and 3	and 3	group3=0
Data collected with success	0.881	0.023	-0.011	0.007	3600	0.184	0.477	0.219	0.512
	[0.324]	(0.023)	(0.014)	(0.013)					
Including:									
Survey completed	0.81	0.009	-0.022	-0.012	3600	0.333	0.473	0.567	0.61
	[0.392]	(0.029)	(0.018)	(0.016)					
Business shut down or owner deceased	0.061	-0.001	0.009	0.004	3600	0.596	0.763	0.644	0.864
	[0.239]	(0.018)	(0.011)	(0.01)					
Small survey on phone for businesses	0.01	0.015	0.002	0.014***	3600	0.184	0.897	0.03**	0.025**
which moved to another location	[0.1]	(0.009)	(0.006)	(0.005)					
Reasons for non-response:									
Refused the survey or not available	0.066	-0.007	0.007	0.009	3600	0.495	0.376	0.836	0.729
	[0.248]	(0.018)	(0.011)	(0.01)					
Moved to another location and not	0.022	0.001	0.004	-0.006	3541	0.749	0.555	0.129	0.473
reached on the phone	[0.146]	(0.011)	(0.007)	(0.006)					
Not found	0.022	-0.012	-0.003	-0.005	3600	0.363	0.45	0.684	0.586
	[0.146]	(0.01)	(0.006)	(0.005)					
Other reason (sickness, traveling during									
the survey, maternity leave)	0.01	-0.005	0.004	-0.006	3600	0.262	0.851	0.021**	0.126
	[0.1]	(0.007)	(0.004)	(0.004)					

Notes: Sources: Survey Data. Column 1: Standard deviations presented in brackets. Columns 2-4: coefficients and standard errors (in parentheses) from an OLS regression of the outcome variable on treatment dummies, controlling for strata dummies (dummies for each triplet). \*\*\*, \*\*, \* indicate statistical significance at 1, 5 and 10%.

#### **Table 5: Program implementation**

		Differe	ence betwe	en []					P-values of joint
	Mean [SD]	and	Control gr	oup		P-valu	e for diffe	rence	test
	in Control					Group 1	Group 1	Group 2	group1=group2=
	Group	Group 1	Group 2	Group 3	Ν	and 2	and 3	and 3	group3=0
PANEL A: Administrative data from CGA a	and Comme	rcial banks	(as of May	2015)					
Step 1: First visit done successfully	0	0.991***	0.973***	0.976***	3600	0.065*	0.085*	0.597	0***
	[0]	(0.009)	(0.006)	(0.005)					
Step 2: Second visit done successfully	0	0	0.416***	0.467***	3299	0***	0***	0.004***	0***
	[0]	(0)	(0.018)	(0.016)					
If 2nd visit conducted, time between	0	0	83.3***	87***	2068	0***	0***	0.353	0***
the two visits (days)	[0]	(0)	(3.4)	(3)					
Step 3: Business is formal acording to the	0	0	0.131***	0.148***	3299	0***	0***	0.179	0***
CGA <sup>β</sup>	[0]	(0)	(0.013)	(0.011)					
Step 4: Additional services:									
Business registered to CGAs	0	0	0.11***	0.125***	3299	0***	0***	0.215	0***
	[0]	(0)	(0.012)	(0.011)					
Business attended to at least one group	0	0	0.058***	0.074***	3299	0***	0***	0.083*	0***
training at CGAs	[0]	(0)	(0.009)	(0.008)					
Business opened an Entreprenant Bank	n.a <sup>α</sup>	n.a <sup>α</sup>	n.a <sup>α</sup>	n.a <sup>α</sup>					
Account									
PANEL B: Midline survey data									
Formalization process: (only formal busine	sses)								
Number of days it took to formalize <sup>n</sup>	26.5	-26**	-20.9**	-16.4**	322	0.655	0.345	0.389	0.052*
	[27.1]	(11.6)	(8.3)	(7)					
Amount paid for formalization	131276	·136062***	102502***	-85927***	318	0.418	0.177	0.393	0.003***
	[162172]	(41852)	(29965)	(25206)					
Share of business who paid nothing for	0	0.882***	0.845***	0.857***	318	0.864	0.897	0.905	0***
formalizing	[0]	(0.218)	(0.156)	(0.131)					
Program Knowledge:		, ,	ι <i>γ</i>	ι, γ					
Ever heard of the Entreprenant status	0.064	0.18***	0.289***	0.3***	2843	0.005***	0.001***	0.628	0***
• • • • • • • • • • • • • • • • • • • •	[0.244]	(0.035)	(0.022)	(0.02)	-				
Ever heard of the Entreprenant status	0.024	0.091***	, 0.191***	0.193***	2843	0.003***	0.001***	0.905	0***
and was able to describe it correctly	[0.153]	(0.03)	(0.019)	(0.017)					

Notes: Column 1: Standard deviations presented in brackets. Columns 2-4: coefficients and standard errors (in parentheses) from an OLS regression of the firm owner/firm characteristic on treatment dummies, controlling for strata dummies (dummies for each triplet). \*\*\*, \*\*, \* indicate statistical significance at 1, 5 and 10%.

 $\alpha$ : Data on who in the study population opened a bank account were not available in time, but we know that 69 businesses in total opened an entreprenant bank account.  $\beta$ : For the control group and group 1, CGA did not have any information as they are not following up with these businesses.  $\eta$ : Top-coded at the 99th percentile.

# Table 6: Impact on Formalization

	(1)	(2)	(3)	(4)	(5)	(6)
Dependent variables:				Showed a document	Showed a	Declared
				or declared formality	document or	formality or
			Showed a	and found in GUFE	found in	found in
	GUFE data	Declared	document	data	GUFE data	GUFE data
Group 1	0.091***	0.041*	0.049**	0.063***	0.100***	0.079***
	(0.021)	(0.024)	(0.020)	(0.022)	(0.025)	(0.027)
Group 2	0.130***	0.112***	0.092***	0.114***	0.149***	0.147***
	(0.013)	(0.015)	(0.013)	(0.014)	(0.016)	(0.017)
Group 3	0.158***	0.122***	0.097***	0.126***	0.167***	0.163***
	(0.012)	(0.013)	(0.011)	(0.012)	(0.014)	(0.015)
Observations	3,600	3,173	3,173	3,173	3,173	3,173
R-squared	0.383	0.446	0.425	0.444	0.442	0.446
Adjusted R-squared	0.073	0.109	0.074	0.105	0.101	0.108
Mean dep. variable in Control	0.009	0.040	0.018	0.018	0.023	0.044
Pvalue Test group1=group2	0.101	0.009	0.058	0.037	0.080	0.022
Pvalue Test group1=group3	0.002	0.001	0.018	0.005	0.008	0.002
Pvalue Test group2=group3	0.032	0.489	0.682	0.422	0.277	0.325

Note: Administrative data from GUFE and survey data March 2015. OLS regression of the outcome variable on treatment dummies, controlling for strata dummies (dummies for each triplet). \*\*\*, \*\*, \* indicate statistical significance at 1, 5 and 10%.

Table 7: Heterogeneous Impact on Formalization by Baseline Characterist	tics
---	------

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
<u>Dependent variables:</u>			For	malized: G	JFE data		
						More than	Female not
		Operates		Species	At least	one visit of	trader
	Female	in Tokpa		classifi-	secondary	a tax	outside
Variable for heterogeneous analysis:	owner	market	Trader	cation	education	inspector	tokpa
Impact in group [] for heterogeneous variable	e=0						
Group1	0.118***	0.106***	0.130***	0.092***	0.058**	0.084***	0.153***
	(0.033)	(0.024)	(0.030)	(0.023)	(0.027)	(0.024)	(0.043)
Group2	0.192***	0.148***	0.170***	0.120***	0.105***	0.122***	0.231***
	(0.022)	(0.015)	(0.020)	(0.014)	(0.016)	(0.015)	(0.029)
Group3	0.220***	0.178***	0.195***	0.144***	0.112***	0.149***	0.230***
	(0.019)	(0.013)	(0.018)	(0.013)	(0.015)	(0.013)	(0.025)
Additional impact in group [] for heterogeneo	ous variable	= <u>1</u>					
Group1 x Heterogeneous variable (int1)	-0.042	-0.069	-0.076*	-0.003	0.080*	0.031	-0.050
	(0.043)	(0.051)	(0.043)	(0.055)	(0.041)	(0.051)	(0.069)
Group2 x Heterogeneous variable (int2)	-0.097***	-0.082**	-0.072***	0.056*	0.064***	0.042	-0.134***
	(0.028)	(0.032)	(0.027)	(0.032)	(0.024)	(0.030)	(0.045)
Group3 x Heterogeneous variable (int3)	-0.097***	-0.089***	-0.067***	0.080***	0.112***	0.049*	-0.078**
	(0.025)	(0.029)	(0.024)	(0.028)	(0.021)	(0.026)	(0.039)
Observations	3,600	3,600	3,600	3,600	3,600	3,600	1,622
R-squared	0.388	0.386	0.386	0.385	0.393	0.384	0.398
Adjusted R-squared	0.080	0.076	0.076	0.076	0.087	0.074	0.088
Mean heterogeneous variable	0.629	0.217	0.549	0.182	0.409	0.196	0.415
Mean dep. var. in Control heterogeneous=0	0.011	0.007	0.007	0.007	0.004	0.007	0.006
Mean dep. var. in Control heterogeneous=1	0.008	0.015	0.011	0.018	0.016	0.017	0.009
<u>Pvalues of test:</u>							
Group1=group2	0.046	0.120	0.242	0.286	0.126	0.158	0.109
Group1=group3	0.002	0.003	0.033	0.028	0.061	0.008	0.072
Group2=group3	0.215	0.045	0.215	0.104	0.720	0.077	0.977
Group1+int1=group2+int2	0.533	0.574	0.187	0.124	0.416	0.365	0.925
Group1+int1=group3+int3	0.096	0.252	0.014	0.010	0.013	0.100	0.355
Group2+int2=group3+int3	0.097	0.413	0.086	0.155	0.017	0.314	0.101

Note: Administrative data from GUFE and survey data March 2015. OLS regression of the outcome variable on treatment dummies and interaction terms (treatment dummies X variable for heterogeneous analysis), controlling for strata dummies (dummies for each triplet). \*\*\*,

\*\*, \* indicate statistical significance at 1, 5 and 10%.

	Mean	Differ	ence betwe	een []					P-values of joint
	[SD] in	an	d Control gi	oup		P-value	e for diffei	rence	test
	Control					Group 1	Group 1	Group 2	group1=group2=
	Group	Group 1	Group 2	Group 3	Ν	and 2	and 3	and 3	group3=0
Legal barriers to formalization									
Business owner has a passport or a Beninese	0.531	-0.028	0.027	0	2877	0.241	0.514	0.305	0.619
ID card	[0.499]	(0.042)	(0.027)	(0.024)					
Business owner has a birth certificate	0.849	-0.029	0.024	-0.007	2877	0.116	0.473	0.104	0.317
	[0.359]	(0.03)	(0.019)	(0.017)					
Business owner has a LEPI card (electoral	0.736	-0.052	0.047**	0.025	2876	0.016**	0.037**	0.349	0.055*
card)	[0.441]	(0.037)	(0.023)	(0.02)					
Perceived costs and disadvantages of formalization	on:								
It increases the amount of taxes to be paid/	0.295	0.052	0.005	0.016	2840	0.284	0.368	0.644	0.595
risk of tax inspection	[0.456]	(0.039)	(0.025)	(0.022)					
High registration costs	0.235	-0.022	-0.024	-0.036*	2840	0.966	0.703	0.598	0.331
	[0.424]	(0.035)	(0.022)	(0.02)					
Registration process complicate or time	0.234	0.003	-0.023	0.006	2840	0.525	0.937	0.214	0.649
consuming	[0.424]	(0.037)	(0.023)	(0.02)					
It involves more corruption	0.025	0.026**	-0.011	-0.005	2840	0.01**	0.019**	0.427	0.071*
	[0.156]	(0.013)	(0.008)	(0.007)					
Other answers	0.042	0.004	0.002	0.01	2840	0.918	0.732	0.47	0.785
	[0.2]	(0.019)	(0.012)	(0.01)					
Don't think there are any disadvantages or	0.199	-0.002	0.043*	0.058***	2840	0.272	0.101	0.508	0.026**
costs	[0.399]	(0.037)	(0.023)	(0.021)					
Don't know	0.236	-0.029	-0.027	-0.058***	2840	0.97	0.392	0.156	0.028**
	[0.425]	(0.035)	(0.022)	(0.019)					
Perceived benefits and advantages of formalizati	on:	(,	(,	(010-0)					
It gives access to new markets or	0.352	-0.037	-0.011	0.011	2841	0.562	0.243	0.41	0.603
government/NGO's programs	[0.478]	(0.041)	(0.026)	(0.023)					
Better reputation/social acceptance for the	0.307	-0.043	0.041	0.006	2841	0.062*	0.223	0.177	0.238
business	[0.462]	(0.04)	(0.025)	(0.023)	-				
Being able to open a bank account/ It is	0.24	-0.004	0.061**	0.073***	2841	0.137	0.05*	0.632	0.004***
easier to get a loan	[0.428]	(0.039)	(0.025)	(0.022)	2011	0.137	0.05	0.032	0.001
It reduces the risk of been ask for bribes/	0 146	-0.001	0.02	0.041**	2841	0 558	0 194	0 308	0 135
being fined /allow to operate more visibly	[0 354]	(0.032)	(0.02)	(0.018)	2011	0.550	0.151	0.500	0.135
Other answers	0 042	0.039**	0.002)	0.009	2841	0 177	0 128	0 974	0.24
	[0 2]	(0.019)	(0.005	(0.005	2041	0.177	0.120	0.574	0.24
Don't think there are any advantages or	0.2	-0.005	0.012	0.002	28/1	0 / / 9	0 500	0.615	0 797
benefits	[0 2/17]	(0.008	(0.010)	(0.000	2041	0.440	0.555	0.013	0.767
Don't know	[0.347] 0.325	0.003	-0.063***	-0.061***	20/1			0.071	0 002***
DONTRIOW	0.225	(0.005	-0.002	-0.001	2041	0.065	0.059	0.971	0.005
	[0.418]	(0.034)	(0.021)	(0.019)					

Notes: Sources: Survey Data. Column 1: Standard deviations presented in brackets. Columns 2-4: coefficients and standard errors (in parentheses) from an OLS regression of the outcome variable on treatment dummies, controlling for strata dummies (dummies for each triplet). \*\*\*, \*\*, \* indicate statistical significance at 1, 5 and 10%.